
CENWP-OP-B                                                                                                                                  24 June, 2002

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 25 of 2002, which covers the period 16 through 22 June 2002.

1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 278.4 to 359.5kcfs.  Daily project tailwater elevation averaged 21.4’ to 25.8’ msl.  Daily project forebay elevation averaged 75.3’ to 75.7’ msl.  River temperature averaged 57 to 58° F.  Daily turbidity measured 3.0’ to 4.5’ on the secchi disk.
1.2 Spill Bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, are operational and included in the spill pattern.  Spill averaged 101.4 to 163.6kcfs for the week.  

1.3 Unit Operation
1.3.1 Unit 1 remains out of service for turbine rehab.

1.3.2 Unit 5 remains out of service for turbine rehab.  

1.3.3 Unit 6 went out of service at 0630 on 17 June for governor modifications.

1.3.4 Unit 7 was out of service from 1330 until 1348 on 18 June for a chip set change on the regulator and from 1330 until 1536 on 21 June due to a tripped breaker.

1.3.5 Unit 8 was out of service from 1306 until 1324 on 18 June for a chip set change on the regulator and from 1330 until 1536 on 21 June due to a tripped breaker.

1.3.6 Unit 9 returned to service at 1636 on 21 June.

1.3.7 Unit 12 was out of service from 0712 until 0742 on 20 June for packing adjustment.

1.3.8 Unit 15 returned to service at 0936 on 18 June.  It had been OOS for VBS repairs.

1.3.9 Unit 17 was out of service for FGE testing from about 1600 until 2100 from 17-21 June.  VBSs were cleaned during swing shift on FGE test nights.

1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.  

1.5 Gatewell drawdowns were measured every day for Unit 17 and on 19 June for the rest of the units.

2.  ACTIVITIES. 
2.1 AWS Closures.  No AWS valves were closed this reporting week.

2.2 STS/VBS Inspections.  The following STS/VBSs were inspected this reporting week.

	Unit
	Date
	Run Hours
	Comments

	1
	6/10
	397
	

	2
	6/19
	804
	

	3
	6/20
	680
	

	4
	6/20
	551
	

	6
	6/19
	785
	

	7
	6/19
	867
	

	9
	6/19
	596
	

	10
	6/20
	745
	

	13
	6/17
	851
	


Table 1.  STS Inspection Dates.

2.3 Dewatering and Fish Salvage.  Biologists removed four 2.5’ sturgeon, two adult lamprey, two sticklebacks, two crawfish, and 12-15 sculpin from Unit 9 tail logs on 19 June.

3.  RESEARCH.

3.1 CRITFC Age and Length Composition.  Collected 100 Spring Chinook and 104 Sockeye this week.

3.2 NMFS FGE.  Unit 17 FGE study is ongoing.  Summer tests are expected to be complete by 12 July.

3.3 NMFS Lamprey.  Collection continues.  Fish are being collected for NMFS, UI, USGS and the Warm Springs tribe.  The lamprey flume will be installed in the Bradford Island AWS channel on 17 June.

3.4 NMFS Saltwater Rearing.  Collection and rearing continue. 

3.5 Normandeau Balloon-tag at Spillway.  Low flow study slated for 19-31 August.

3.6 OCFWRU Transportation Study.  Study of post-transportation migration and survival of juveniles to run 29 April through 14 July.

3.7 ODFW Summer Steelhead sampling.  Sampling 2 days/week, 8 hours/day.  Will sample 1% of the run.

3.8 Pelfrey Sturgeon Hatchery.  Capturing broodstock May 24 – June 24.

3.9 PSMFC Adult PIT.  Detectors are installed in all ladders.  Some of the detectors have developed leaks and will be monitored throughout the season.  Repairs are scheduled for 2002-03 winter maintenance season.

3.10 PSMFC JMF.  Two kelts were released alive from the separator bars.  92 fallbacks (67 steelhead, 20 salmon and 5 unknown salmonids) and 225 smolts were observed sliding over the separator bars.

3.11 PSMFC/NMFS Juvenile PIT.  200 sub yearling Chinook were examined for signs of GBT.  Total descaling for all fish sampled at PH1 was 4.9%, down from 6.7% last week.  On 20 June the south gate opened for no apparent reason.  Operators restored the system to normal within 25 minutes (see misc. 5.5).

3.12 U of I Adult Radio tracking.  Trapping began 1 April and will continue 7 days/week until October.

3.13 U of I Transportation Study.  Evaluation of physiological condition of river run v. barged Snake River juveniles to occur June-September.

3.14 USGS Juvenile Radio tracking.  Study in progress.

3.15 USGS Lamprey.  Collection continues.

3.16 USGS Sort by Code.  Length and weight are recorded and fish are released back to the river.

3.17 WDFW Fish Counting.  16-hr visual and 8-hr video counts at Bradford Island and Washington Shore.

3.18 WES/Battelle/Mevatech Juvenile Hydroacoustics.  Study in progress.  

3.19 WES Spillway Deflector Testing.  WES monitoring water quality equipment to determine effectiveness of new flip lips.  Study began with initiation of spill on 10 April.

4.  FISHWAYS 
4.1 Project biologists inspected 16-18, 20-21 June this reporting week.  NMFS inspected on 19 June.

4.2 Fish Passage Plan Observations.

	DATE
	LOCATion
	FPP Deviation
	CAUSE
	RESPONSE
	TR/WAR

	All week
	PH1 PCC velocity
	Velocity <1.5fps on the electronic meter.
	Unknown- the rest of the ladder is in criteria.
	No action at this time.
	

	6/19, 6/22
	
	Biologists did not inspect 7 days/ week.
	Personnel shortage.
	Job advertised.
	

	6/19
	B-Branch/ Cascade Island
	Diffusers not in correct position for TW elevation.
	Computer system not yet fully functional.  See misc. 5.1
	Operators manually adjusted diffusers.
	

	6/20
	PH1 PCC
	OG#9 and OG#62 too low for TW elevation.
	Transducers need to be replaced.
	Gate manually adjusted.
	

	6/21
	Weir 37
	Reading <1.3’
	Unknown
	Operators adjusted the control.
	


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.  Temperatures are in Rankine this week.  Rankine is an absolute temperature scale named after Scottish engineer and physicist William John Macquorn Rankine, one of the founders of thermodynamic theory.  http://students.washington.edu/kyle/temptable.jpg
	Date
	Wash. Shore
	Bradford Is.

	6/16
	519.21
	519.21

	6/17
	519.21
	519.03

	6/18
	519.39
	519.39

	6/19
	489.75
	489.75

	6/20
	490.29
	490.29

	6/21
	520.83
	520.83

	6/22
	521.37
	521.37


4.4 Sensor Calibration.  Calibration was checked on 21 June this reporting week.  

4.5 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week. 

5. MISCELLANEOUS

5.1 B-branch and Cascade Island are operated in manual mode until stillwells can be installed for the tailwater sensors.  Installation will wait until spill has concluded.

5.2 Turbulent flows from spillbays 1 and 18 enter the side entrance gates of Cascade Island and Bradford Island B-branch ladders, causing poor entrance conditions.  District hydraulic engineers are investigating the problem.  On 3 May, the structural crew installed a bulkhead in the side entrance at Cascade Island.  Entrance conditions improved.  CG funds will be used to build a bulkhead for Bradford Island B-branch.

5.3 There has been some difficulty in maintaining an opening for fish at PH1 weir gate 1.  The gate is currently programmed not to close if there is less than an 8’ opening, but it also won’t open automatically unless there is a problem with differential.  Electricians are exploring the possibility of reprogramming the gate such that it will open if there is less than a 6’ opening and not close if less than 8’.  Until then, both WG1 and FV1-1 will be manually adjusted as necessary to keep an opening at the south end of PH1 collection channel.

5.4 On 20 June, dive crews seined the old navlock in preparation for the closing of the miter gates.  After almost a decade of inoperation, the miter gates silently and smoothly swung closed at 1307. 

5.5 From 1125 until 1150 on 20 June, the South Gate at DSM1 opened.  The cause is yet unknown.  Six to eight smolts were reported as stranded on the inclined screen before operators were able to restore the system to normal.  No fish were reported as morts.

5.6 The brusharm chair for Screen Cleaner #1 broke at 0510 on 21 June.  PH2 mechanics had it repaired and operable by 0808 that same day.

 DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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